

# Alma Mater Studiorum Università di Bologna Archivio istituzionale della ricerca

Facebook as a Destination Marketing Tool: Evidence from Italian regional Destination Management Organizations

This is the final peer-reviewed author's accepted manuscript (postprint) of the following publication:

Published Version: Facebook as a Destination Marketing Tool: Evidence from Italian regional Destination Management Organizations / Mariani, MARCELLO MARIA; DI FELICE, Marco; Mura, Matteo. - In: TOURISM MANAGEMENT. - ISSN 0261-5177. - ELETTRONICO. - 54:(2016), pp. 321-343. [10.1016/j.tourman.2015.12.008]

Availability: This version is available at: https://hdl.handle.net/11585/533553 since: 2022-02-01

Published:

DOI: http://doi.org/10.1016/j.tourman.2015.12.008

Terms of use:

Some rights reserved. The terms and conditions for the reuse of this version of the manuscript are specified in the publishing policy. For all terms of use and more information see the publisher's website.

This item was downloaded from IRIS Università di Bologna (https://cris.unibo.it/). When citing, please refer to the published version.

(Article begins on next page)

# This is the final peer-reviewed accepted manuscript of:

Mariani, M. M., Di Felice, M., & Mura, M. (2016). Facebook as a destination marketing tool: Evidence from Italian regional Destination Management Organizations. *Tourism management*, *54*, 321-343

The final published version is available online at:

https://doi.org/10.1016/j.tourman.2015.12.008

Rights / License:

The terms and conditions for the reuse of this version of the manuscript are specified in the publishing policy. For all terms of use and more information see the publisher's website.

# Facebook as a Destination Marketing Tool: Evidence from Italian regional Destination Management Organisations

# Abstract

This work explores how Italian regional Destination Management Organisations (DMOs) strategically employ Facebook to promote and market their destinations, and improves on the current metrics for capturing user engagement. Based on big data analysis from the regional DMOs' Facebook pages, supplemented with semi-structured interviews conducted with DMO managers, the study sheds light on the factors contributing to superior level of social activity. The findings indicate that the way Facebook is tactically and strategically employed varies significantly across Italian regional DMOs. Visual content (namely photos) and moderately long posts have a statistically-significant positive impact on DMOs' Facebook engagement, whereas high post frequency, and early daily timing (in the morning) of posts have a negative impact on engagement. Last but not least, the study shows that most of the regional DMOs (except for Trentino, Tuscany, and Sicily) deploy Facebook with a top-down approach, allowing for little spontaneous user generated content (UGC).

Key words: Social media, Facebook, Big Data, Destination Management Organizations, engagement, destination marketing, Italian regions

# 1. Introduction

Over the last 30 years the impact of Information and Communication Technologies (ICTs) on the global economy has been dramatic, completely changing the way human activities are performed, material and immaterial resources are mobilised, wealth is generated, and business opportunities are created and seized. Among the many industries affected by technological advances, tourism is undoubtedly one where ICT-driven business process re-engineering has engendered a new paradigm shift, because: "not only do ICTs empower consumers to identify, customise and purchase tourism products but they also support the globalisation of the industry by providing effective tools for suppliers to develop, manage and distribute their offerings worldwide" (Buhalis and Law, 2008).

The Internet has deeply transformed the manner in which travellers access information, plan for and book trips, and subsequently share their travel experiences (Buhalis and Law, 2008; Chung and Buhalis, 2008; Hays et al., 2013; Xiang and Gretzel, 2010; Bilgihan et al., 2016). In the early 'Web 1.0' days of the Internet, information published online was static, and interaction with other Internet users and/or publishers almost non-existent (Hays et al., 2013). By contrast, the 'Web 2.0' Internet of today is characterised by review sites (e.g., Tripadvisor.com) and social media platforms (such as Facebook.com, Twitter.com) that have brought about a social-network revolution of the World Wide Web, leading to what has been termed 'socialnomics', i.e., an economy revolving around social media (Qualman, 2009). In fact, social media websites enable Internet users to form virtual travel communities (VTCs) that make it easier for tourists to obtain information, maintain connections, develop relationships, and eventually make travel-related decisions (Ayeh, Norman & Law, 2013; Stepchenkova et al., 2007; Wang et al., 2002).

Social media sites facilitate consumer-generated content (CGC) such as blogs, photos, videos, wikis, or reviews (Boyd and Ellison, 2008; Sigala et al., 2012; Wenger, 2008) and are widely used by online travellers (White and White, 2007), thereby marking a shift toward users rather than organisations taking charge of Internet content (Schegg et al. 2008). Sometimes CGC and online reviews (such as those published by Tripadvisor.com) might even challenge the authority of established Destination Management Organizations (DMOs) and conventional advertisements, by undermining their reputation (Dijkmans et al., 2015; Gretzel, 2006; Gretzel et al., 2000). Consequently, DMOs at both the national and local level need to pay attention to how they use social media, as this can play a crucial role in effectively promoting and marketing a tourism destination in a global context, characterised by hyper-competition not only among companies (D'Aveni, 1994) but also among destinations.

This work contributes to the extant body of 'e-tourism research'—that is, research at the intersection between tourism and ICTs-- from a number of perspectives. First, it adds to the currently scant stream

of empirically-based studies that investigate DMOs' social media marketing from both a qualitative and a quantitative perspective (Hays et al., 2013). Specifically, it explores how Italian regional DMOs employ Web 2.0--and specifically Facebook--to promote and market their destinations, illustrating their varying degrees of social-media usage. Second, this work advances the currently available quantitative methodologies for capturing Facebook-user engagement: it addresses the potential issues and biases inherent in existing engagement metrics, and employs a novel software tool, specifically developed for this study, to retrieve data from Facebook pages. Third, this work innovatively identifies and measures the factors that contribute to increased levels of social activity--also using qualitative data to help interpret the quantitative findings. Finally, by triangulating the quantitative and qualitative methods, several managerial best practices are identified that DMOs can adopt in their use of social media. The paper is organised as follows. Section 2 gives a review of relevant literature on e-tourism research and social media. Section 3 describes the methodology that was employed, while Section 4 reports the research findings and the analysis. Finally, Section 5 discusses both the managerial and policy-making implications of this study, sets out its limitations, and maps out avenues for further research.

#### 2. E-tourism research and social media

Research on the role of ICTs in the tourism sector (commonly referred to as e-tourism research) has consolidated significantly over the last three decades, mirroring the growing number of ICT applications in the sector and the acknowledgment of the tourism industry as a leading e-commerce application (Wethner and Ricci, 2004).

Buhalis and Law (2008), in a survey of the academic articles published in top-impact academic journals, identify three major axes within e-tourism research: (1) technological innovation; (2) industry functions; (3) consumer and demand dimension. With respect to technological innovation (the first axis), in recent years a number of technologies have been identified as critical for further innovation in the tourism industry. They are related mainly to interoperability and ontology-building (Stabb and Werthner, 2002; Jakkilinki et al., 2007; Baggio, 2014), multimedia (Fiore et al., 2005, Raggam and Almer, 2005), mobile and wireless technologies (Flouri and Buhalis, 2004; Alfaro et al., 2005), wireless local area networks (WLANs) and the worldwide interoperability for microwave access (WiMax) (Odinma et al., 2007), location-based services (LBS) (Berger et al., 2003), web design in both functionality and usability (Hashim et al., 2007; Baloglu and Pekan, 2006), accessibility (Michopoulou et al., 2007), and ambient intelligence (AI) (Buhalis and O'Connor, 2005; Yovcheva et al., 2012).

The industry functions (second axis of e-tourism research) most strongly impacted by the aforementioned technologies are marketing and distribution (Go and Williams, 1993; O'Connor and Frew, 2002; Buhalis and Licata, 2002), because by helping to promote mass-customisation, these

technologies have made it possible to target niche markets of significant size in different geographical locations.

Finally, the customer and demand dimension (third axis of e-tourism research) has to do with markets and customers. This axis covers the idea that ICTs offer a range of tools to facilitate and improve the entire process, starting from information search, through to destination/product consumption, and post-experience engagement, (Gursoy and McCleary, 2004; Luo et al., 2004; Frias et al., 2011; Borges, 2009; Parra-Lopez et al., 2011). Such tools comprise pricing devices (Clemons et al., 2002), recommender systems (Fesenmeir et al., 2003; Ricci and Wethner, 2006), profiling methods (Wang et al., 2004; Tufte and Rasmussen, 2002; Matloka and Buhalis, 2010), mechanisms for dealing with complaints and electronic word-of-mouth (eWOM) (Gelb and Sundaram, 2002; Shea et al., 2004), customer-centric marketing (CCM) systems (Niininen et al., 2007), social media and VTCs (Stepchenkova et al. 2007; Wang et al., 2002; Vogt and Fesenmaier, 1998; Xiang and Gretzel, 2010; Bras et al., 2010).

With the emergence of Web 2.0, the concept of social networking/virtual communities applied to the tourism sector has resulted in 'Travel 2.0'--considered to be one of the "mega trends" to most significantly impact the tourism system (Leung et al., 2013). More specifically, a "virtual community is a group of people who may or may not meet one another face-to-face, and who exchange words and ideas through the mediation of computer bulletin boards and networks" (Rheingold, 1993: p. 58). A VTC is simply a tourism-themed virtual community, in which tourism-related CGC is created (Kaplan and Haenlein, 2009). TripAdvisor (www.tripadvisor.com) is one of the most successful examples of a virtual community in tourism: it facilitates the reviewing of hotels, restaurants, and destinations around the world and brings together individuals in discussion forums. The system provides users with independent travel reviews and comments written by TripAdvisor members and expert advisors, and represents a powerful platform for interaction among peers (Wang and Fesenmaier, 2004).

The recent tourism and hospitality literature has addressed the role and use of social media in travellers' decision-making, as well as in tourism operations and management (Sigala et al., 2012). While consumer-centric studies have generally focused on the use and impact of social media in the research phase of travellers' planning process (Ayeh, Norman & Law, 2013), supplier-related studies have concentrated closely on the promotion, management, and research functions (Leung et al., 2013).

Social media is becoming more and more relevant as a destination marketing tool, and accordingly must be effectively managed by DMOs looking to create a sustainable competitive advantage for their destinations within the competitive arena of global tourism (Pike and Page, 2014). This is all the more relevant in the current economic climate of public austerity, characterised by a decline in funding for both National Tourism Organisations (NTOs) and local DMOs. This state of

affairs has pushed many DMOs, albeit sometimes slowly, to turn to social media as a relatively low-cost marketing tool with a global reach. Notwithstanding this trend, research on the use of Web 2.0 and social media as destination marketing tools remains scant (Feng et al., 2003; Stankov et al., 2010; Xiang and Gretzel, 2010; Hays et al., 2013).

To help fill this research gap, we carried out a study with the following four main objectives: (1) to explore how Italian regional DMOs strategically employ Facebook to market their destinations; (2) to illustrate the varying degrees of Facebook usage among the studied DMOs; (3) to determine what factors contribute to increased engagement of tourists/users through a statistical analysis; (4) to identify several managerial best practices for the use of Facebook by Italian DMOs.

We elected to focus on Facebook, rather than on other social media (such as Twitter), because the creation and adoption of Twitter accounts on the part of Italian regional DMOs is still in its infancy, whereas a preliminary pilot study showed that Facebook is (as of November 2014) the most-used social media platform among Italian regional DMOs.

# 3. Methodology

This study investigates the nature and the degree of Facebook usage by regional DMOs in Italy. Following Hays et al. (2013), the research is mainly exploratory in its aim, and accordingly adopts a mixed-method approach that employs both qualitative and quantitative data.

As a pilot study, we conducted preliminary interviews with the social media managers of three major Italian DMOs (Emilia-Romagna, Tuscany, and Marche). The interviews were "informal, conversational interviews" that relied chiefly on the interaction between interviewer and respondent (Patton, 1990). The evidence obtained from those interviews was triangulated with data from publicly available documents (Gioia et al., 2013). This preliminary research highlighted a number of issues, especially relating to how major destination marketers use and make sense of social media. In fact, although Italian regional DMOs employ various social media platforms (such as Twitter, Youtube, Google Plus, Instagram, Flickr, Pinterest, Panoramio, My Space, Issuu, Foursquare, and Tumblr) Facebook--with millions of users worldwide and 18.3 million active users in Italy by the end of 2014--appears to be the one most widely adopted by Italian DMOs.

In the light of those pilot interviews and the extant literature in the field, we selected Facebook as the data source for this work and accordingly collected data from the Facebook pages of all the Italian regional DMOs. Specifically, we gathered information on the number and types of posts published on DMOs' Facebook pages, and on the user responses elicited by these posts. The data retrieval, aggregation, and analysis were performed using a novel software tool that we developed specifically for this purpose, and whose main features are presented in Section 3.2.

Facebook content is freely available on the web, and can be regarded as sufficient for this study from an empirical point of view (Hays et al., 2013). The collected data were analysed using descriptive statistics, and multivariate regression analyses were employed to explore the main determinants of users' engagement in social media activity.

After gathering and analysing the data, we carried out semi-structured interviews with social media marketers from seventeen Italian regions, in order to investigate the issues that emerged from the quantitative analysis in more depth and strengthen the validity of the findings. To capture a more complete picture of DMOs' social media activity, interviewees from DMOs were chosen on the basis of theoretical sampling (Yin, 1994), following the characteristics that had emerged as most important in the quantitative data analysis. During this stage, at least two interviewers were present at each interview. When recording was possible (the DMOs were sometimes reluctant for us to do so), the material was subsequently coded independently by each researcher (using the NVivo software) and then compared. The analysis of both the qualitative and quantitative data acquired in this way produced a series of findings that are presented and discussed in Section 4.

# 3.1 Relevance of the context

Italy and its regional destinations are among the most-visited destinations worldwide, with Italy ranking 5<sup>th</sup> in terms of international tourist arrivals worldwide (UNWTO, 2015), and with domestic arrivals exceeding international ones by 5 million (ISTAT, 2015). While the promotion of Italy as a country destination is handled by the Italian National Tourism Organisation ENIT (Ente Nazionale Italiano Turismo), which has only recently begun to develop a social media strategy<sup>1</sup>, the leading role in destination marketing is played by the individual regional DMOs, which by law have a mandate to promote their tourism destinations. In Italy, the main responsibility for tourism promotion has traditionally lain with the regions rather than with the central government: Law 217 of 1983 established the "Aziende di Promozione Turistica" (APT)—that is, regional DMOs--tasked with promoting the regional destinations and providing information for tourists. For this reasons, regions (and their DMOs) in Italy have a much bigger budget (in many cases several times higher) than the national organisation, ENIT. What's more, regional DMOs individually allocate more financial resources to digital marketing than does ENIT. Consequently, the investigation focused on all the Italian regions (twenty in total), and more specifically on the DMO of each region.

Italy has 20 regions, and most of the Italian regional DMOs have their own official Facebook page and

<sup>&</sup>lt;sup>1</sup> Interestingly, Italy crafted its first Strategic Plan for Tourism (Turismo Italia 2020) only in 2013. However, that plan was never approved by the Ministry of Tourism since then, and the Italian NTO has been present on Facebook since 2011.

provide original content to promote their destinations and enhance user engagement. Although every Italian region has an official website, not all of them have a well-developed social media presence. In all the cases analysed here, the DMO's Facebook page was linked to the official website of the DMO (and vice versa).

Since not all regional DMOs have developed Facebook pages in foreign languages, and since the pages in Italian have far more fans and social activity than their foreign counterparts, the analysis was carried out on the Italian-language pages--which by definition target domestic tourists. While this might seem a limitation, it has the advantage of ensuring consistency for the analysis and enables us to make valid comparisons across regions. Many regional DMOs have adopted social media platforms as part of their marketing efforts--mostly during the last five years, and so significantly lagging behind many of their international English-speaking counterparts. The Valle D'Aosta DMO's Facebook page was the very first one to open, in May 2009, and a record number of DMO Facebook accounts (eight) were created in 2011. By December 31, 2013, all Italian regional DMOs apart from one--Lazio--had a Facebook page publicly visible on the Internet<sup>2</sup>.

#### 3.2 Data gathering first phase: Facebook data

We collected data from the official Facebook pages of all Italian regional DMOs that had such a page (19 out of 20 regions did). This was done using a novel software tool that we developed specifically for this study, based upon cutting-edge technologies for managing data retrieved from the Facebook platform. The architecture of this tool, designed by one of the co-authors (who is an information scientist), is depicted in Figure 1.

Add Figure 1 about here

\_\_\_\_\_

The data-acquisition tool is composed of four main software modules, acting on a pipeline: (i) An extractor module, which retrieves the data publicly available on the DMOs' Facebook pages, through data queries based on the Graph API technology (Facebook Graph API 2.0, 2014); (ii) A parser module (written in Python language), which processes the results of each query (in JSON language format), and computes per-post statistics and reports; (iii) An analyser module, which computes aggregated

<sup>&</sup>lt;sup>2</sup> The Lazio region maintained only an intermittent online presence during the analysed period owing to a number of political issues. As a consequence, its Facebook page was not visible.

engagement metrics over a time window defined by the user (in our case, one month), and (iv) A data visualisation module, which displays the results through graphs, and allows exporting reports in .CSV data format. Although there are several general-purpose software packages available on the market that provide social-media analytics functionality, this tool provides some unique features that justify its novel deployment from scratch. Specifically: (*i*) it is natively *multi-metric*, i.e. it is able to compute and return to researchers a large variety of engagement metrics (some of these are reported in Table 2 and Table 3), to provide a comprehensive understanding of social media activities from a variety of perspectives, (*ii*) it is highly *modular*, i.e. the pipeline of Figure 1 can be easily extended with additional block components without modifying the core architecture of the tool; in fact, we are currently working on a new component for performing semantic analysis of extracted posts, that acts on the output of the analyser module, (*iii*) it can be customised for different types of social-media analysis, by appropriately tuning a large set of parameters that control both the data extraction and data aggregation stages (e.g. time window of the analysis, time granularity, contents to process) through its graphical user interface (GUI). It should also be noted that the capabilities of this tool, as developed thus far, are already sufficiently powerful to justify its use also for other social-media related analysis.

For this study, we extracted the overall population of posts (including the related 'Likes', 'Shares' and 'Comments') present on the Facebook pages of Italian regional DMOs during the year 2013 (from the 1st of January to the 31st of December 2013), obtaining a total of 33,597 posts<sup>3</sup>.

The contents of those posts were analysed using measures adapted from Hays et al. (2013), as well as six additional measures which we developed specifically for this research (see Table 1 for details).

Add Table 1 about here

Additionally, based on Frick (2010), we calculated a number of metrics to assess engagement: generic engagement, brand engagement, and user engagement (see Table 2 for details).

Add Table 2 about here

<sup>&</sup>lt;sup>3</sup> All the collected data were publicly available on the Facebook pages of the DMOs.

We also computed those same measures of engagement on a rolling basis, in a two- and seven-day time window after the posting date. These rolling metrics were based solely on the 'Comments' and did not include 'Likes' and 'Shares', for which we did not have an exact timestamp (provided by the combination of date and exact hour/minute of the liking/sharing event). Because they also capture the longitudinal dimension, these rolling measures of engagement are an advance over the existing empirical methods that have sought to gauge user engagement with social media in the tourism sector (Hays et al., 2013; Oviedo et al., 2014).

Finally, we included three additional measures of engagement which only take into account user activity, without normalising it by the number of fans (see Table 3 for details). In so doing, we sought to overcome one of the chief shortcomings of the normalised engagement measures usually adopted in the literature (Oviedo et al., 2014), whose value over time tends to decrease as the number of fans increases, because they include the number of fans in the denominator. These new measures are in line with recent methodological developments in the marketing literature (Oviedo et al., 2014) and also with the recommendations issued by Socialbakers--the most popular provider of social-media analytic tools--which suggest that normalising by fan size does not yield reliable engagement metrics when public data are extracted over a long period of time with benchmarking objectives, "because the size of the community that sees promoted posts is flexible" (Socialbakers, 2015).

Add Table 3 about here

-----

The aforementioned metrics of engagement were calculated by applying different weightings to Likes(weight=20%), Shares (weight=30%) and Comments (weight=50%). These differential weights are a proxy for the degree of involvement implied by the underlying activities of respectively liking, sharing, or commenting. The selected setting of the weights (20-30-50) is based on an important antecedent of users' activity (the time required to complete the action) and on an outcome (the visibility and effectiveness of dissemination of the users' actions and content).

Commenting on a post implies that a user has invested time in writing and sharing an opinion (Sabate et al., 2014): writing a comment will typically take more time and effort than merely sharing or liking a post, with the two latter actions being simple binary decisions, accomplished by a single click. Consequently, we hypothesise that 'commenting' is associated with higher levels of user involvement than 'sharing' or 'liking'. Furthermore, 'sharing' is weighted more than 'liking' because it leads to more effective dissemination of the message to other potential users through the Facebook algorithm (Moore

and McElroy, 2012; Ruiz-Mafe et al., 2014). It is important to note that this weighting scheme is an improvement over those used by previous empirical studies for gauging user engagement for social media marketing (Oviedo et al., 2014). We also conducted a robustness analysis of this new scheme by using different weights. Although changing the weight percentages yielded different absolute values of the engagement variable, the resultant ranking of DMO Facebook pages in terms of engagement did not change (as we will show in Section 4.1) and so the weightings did not have an impact on the analysis of the determinants of engagement (Section 4.2). In other words, the results presented in this study are independent of the weighting scheme adopted for computing the engagement metric.

# 3.3 Data gathering second phase: semi-structured interviews

Based on the pilot interviews and on publicly available documents, we prepared a list of information needs and interview instructions, as well as a list of open-ended questions that we used for conducting semi-structured interviews (see Table 4).

Add Table 4 about here

Semi-structured interviews were conducted with social media marketers of seventeen regions: Abruzzo, Basilicata, Calabria, Emilia-Romagna, Friuli Venezia-Giulia, Liguria, Lombardia, Marche, Piedmont, Puglia, Sicily, Sardinia, Trentino, Tuscany, Umbria, Valle d'Aosta, and Veneto<sup>4</sup>. The information from those interviews served to complement the data obtained from quantitative research, to yield a fuller understanding of how Italian regional DMOs implement and deploy their social media strategies. The researchers conducting the interviews posed open-ended questions and took notes of the responses.

#### 3.4 Data triangulation

The field note transcripts from the semi-structured interviews were independently coded by two of the three researchers using NVIVO, to separately identify terms and patterns in the data. Overall, a number of themes emerged from this process (e.g., interaction of DMOs with Facebook users, topics and themes of social activity, audience development, staff and training related to social media and Facebook, tools to assess the effectiveness of Facebook use) that provided a deeper understanding of

<sup>&</sup>lt;sup>4</sup> Interviewees included social media managers, social media strategists, community managers, and digital strategists of the aforementioned regions.

the results of the quantitative analysis. The coding process was carried out adopting an iterative approach of data comparison and contrast (Miles and Huberman, 1994), and considering the extant literature on social media marketing in tourism.

Combining the quantitative results with this interview data made it possible to refine the analysis (Jick, 1979). In fact, the triangulation of data from multiple sources reduces construct validity problems and the risk of retrospectively imposing meaning on historical events based on knowledge of the outcomes. By leveraging the qualitative evidence collected through the interviews, we were able to further validate the descriptions and interpretations of the quantitative analysis (Miles & Huberman, 1984). All the interviewees were also invited to read a preliminary draft of this paper, and provide their comments and feedback.

# 4. Analysis and results

During 2013, the Facebook pages of Italian regional DMOs recorded a total of 33,597 posts, of which 18,046 (53.7%) were posts made by the DMOs themselves (Table 5). This percentage suggests that the goal of two-way communication between DMO managers and fans has been achieved, given that half the posts come from users. The clear-cut leader in terms of total posts was the Trentino DMO page, followed by Emilia-Romagna and Marche. Trentino also had a very high percentage of posts by users<sup>5</sup>, as did Tuscany and Sicily.

Add Table 5 about here

\_\_\_\_\_

Overall there does not appear to be any stark 'social media divide' between the north and south of Italy. However, it is true that the top five DMOs in terms of total Facebook posts (Trentino, Emilia-Romagna, Marche, Liguria and Tuscany) are concentrated in northern and central Italy (Figure 2).

-----

Add Figures 2 and 3 about here

-----

If we now look only at the posting activity of the DMOs themselves (excluding that of users), the general results do not change much, except that the leaders now are Trentino, Emilia-Romagna, Marche, Piedmont, and Liguria (Figure 3). The regions with the highest percentage of posts by users are

<sup>&</sup>lt;sup>5</sup> The figure for Campania is not relevant due to the very limited number of total posts

instead Campania<sup>6</sup>, Sicily, Trentino, and Tuscany. Setting aside Campania (owing to its very low total posts), this means that Sicily, Trentino, and Tuscany are the regions whose users post most actively.

Most of the content posted by DMOs includes photos (81.2% on average for the overall population in 2013), followed by links (10.4%) and videos (5.9%). The content-type breakdown for each regional DMO is presented in Figure 4:

-----

Add Figure 4 about here

-----

Campania, Emilia-Romagna, Tuscany, Trentino, Veneto, and Sardinia are the regions with the strongest preference for posting photos, as well as being among those that post most actively in general. If we exclude the outliers Molise and Campania (which have a very low number of total posts), the cases of Sicily and Piedmont seem to stand out from the overall picture, mostly due to their extensively use of links for highlighting events.

Turning now to focus on user activity (see Table 6), we find that Trentino, Tuscany, Marche, Sicily, and Liguria are the leading DMOs in terms of Likes (with 712,071; 607,644; 309,767; 268,439; and 259,598 Likes respectively) while at the bottom of this ranking are Campania, Valle D'Aosta, and Molise (with 872, 844 and 0 Likes respectively, and so well below the psychological threshold of 1000 Likes, which is instead surpassed by the fourth-from-last region, Piedmont, with 1921 Likes). The top 5 DMOs in terms of Likes are also those that lead the ranking for Comments (Trentino, Tuscany, Sicily, Liguria, and Marche with 81,341; 30,929; 19,508; 15,688; and 12,032 Comments respectively), while at the bottom of this ranking we again find Campania, Valle D'Aosta, and Molise (with 30, 24, and 0 Comments respectively). Finally, the top 5 DMOs in terms of Likes and Comments are also those that lead the ranking for number of Shares (Trentino, Tuscany, Marche, Sicily, and Liguria with 215,297; 172,619; 144,706; 134,622; and 70,701 Shares respectively), while among the worst performers we again find Valle D'Aosta, Campania, and Molise (with 34, 0, and 0 Shares respectively).

Examining the last 6 columns of Table 6, it is clear that the majority of Likes, Comments and Shares are generated by users. Overall, the top 5 regional DMOs in terms of user

<sup>&</sup>lt;sup>6</sup> Again, the figure for the Campania region is not relevant

activity are also the top 5 regional DMOs in terms of posts. Therefore, it appears that high posting activity (in terms of number of posts) on a DMO page coincides with high user activity.

With a few exceptions (see Umbria, Basilicata, and Lombardy in terms of the proportion of comments made by users), most of the Likes, Shares and Comments originate from users (i.e., the DMO engages in posting activity but is rarely involved in liking/sharing/commenting activities).

Interestingly, although Trentino and Emilia-Romagna have a similar number of posts made by the DMO (see Table 5), the Trentino users tend to interact more (by commenting, liking, and sharing more items). This may be ascribable to the higher loyalty of Trentino users, resulting from that DMO's adoption of managerial best practices (as will be discussed further in Section 4.3. "Interaction with consumers and users"). Figure 5 visually illustrates the situation across DMOs.

Add Table 6 about here ------Add Figure 5 about here

The regions that had the highest number of fans at the end of 2013 are mostly the same as those with the highest numbers of Likes, Shares, and Comments--with the exception of Puglia which, despite modest user activity, had a strong rise in its following to reach 60,246 fans. However, as is clear from Table 7, the largest percentage increases over the course of 2013 occurred for DMOs of smaller regions such as Umbria (+7429%) and Calabria (+1925%)--either because they were the ones to most recently open a Facebook page, or because their original fan base on January 1<sup>st</sup> 2013 was very thin.

Add Table 7 about here

-----

4.1 Active users and engagement metrics

The descriptive statistics reported thus far do not include a proper assessment of user activity. In fact, the users totalled above may be active or inactive. Figure 6 shows that the number of active users (defined as those who posted at least one comment in 2013) is far less than the total number of fans for each DMO page. This suggests that the fan base of a DMO page is not a reliable normalisation variable to be used for computing engagement metrics.

Add Figure 6 about here

Overall, the five regions most active in posting (except for Emilia-Romagna) are also those with the highest number of active users.

Looking more in depth at the distribution of the number of comments made by active users, a significant finding emerges-- on average, almost 95% of active users (i.e., users who posted at least 1 comment during the year), posted less than 5 comments in total during 2013 (Figure 7).

Add Figure 7 about here

Trentino shows the highest percentage of 'repeat users'--meaning users who intervened more than once in threads initiated by the DMO. This might partially explain why the Trentino DMO has more comments than any other DMO (see Table 6). Technically, we can say that the active users of the Trentino DMO page show stronger loyalty than those of other regional DMOs. The reasons for this are explained in Section 4.3.

Turning now to focus on the engagement metrics, we find that for generic and brand monthly engagement metrics (Figure 8), regions such as Calabria, Umbria, and Veneto display the highest values. This seems to contradict the preceding results for user-generated activity in terms of Likes, Shares, and Comments. However, it can be explained by noting the construction of the metrics, which incorporate the overall number of fans and posts (respectively "total" and "by the DMO") into the denominator of the ratio—thereby favouring those regions that have few fans.

\_\_\_\_\_

# Add Figure 8 about here

-----

If we assess the level of activity linked to each individual post and do not normalise it by the fan base (but only by the posts by users), as suggested in the most recent literature (Oviedo et al., 2014), the results of the analysis change completely: Trentino, Tuscany, Marche, Sicily, Liguria instead become the top performers in terms of engagement (Figure 9).

\_\_\_\_\_

Add Figure 9 about here

These engagement results seem a more adequate proxy for capturing the actual involvement of users, and are consistent with the results shown in Table 6 and Figure 5.

If we refine the metric even further, so that engagement is normalised neither by fan base nor by number of posts, the results become closer to those of the preceding figure (Figure 9) and more consistent with those obtained from Table 6 and Figure 5:

Add Figure 10 about here

The DMOs with the highest values of 'user' and 'user-to-DMO' interaction are Trentino, Tuscany, Marche, Sicily, and Liguria (see Figure 10). Interestingly, 'user-to-user' interaction is instead extremely low, suggesting that posts written by users do not attract the attention of other users, often because they are in plain text (they do not include pictures, videos, etc.). This result also seems to indicate that Italian regions still deploy social media in a traditional top-down fashion, rather than exploring ways to foster a bottom-up UGC-driven approach. This might simply be due to the fact that UGC is on average less interesting and 'professional' than content generated by the DMO.

As mentioned in the methodology section, the engagement metrics were computed applying an innovative scheme to differentially weight Comments, Shares, and Likes. To ensure methodological accuracy, we conducted a robustness check by using different weights. Figure 11 shows the Generic Engagement for Users' Activity (the yellow bar in Figure 9) obtained by using three different tunings of the weight factors, defined as follows: Setting1 (Comment=50%, Share=30%, Like=20%), Setting2 (Commens=20%, Share=30%, Like=50%) and Setting3 (Comment=20%, Share=50%, Like=30%). Figure 12 shows the rankings of the DMO Facebook pages according to the engagement metric

computed with each setting. Comparing the results of Figures 11 and 12, we can conclude that different weight settings generate different absolute values of user engagement, but do not alter the ranking of DMO Facebook pages in terms of engagement. Therefore, the new engagement variable has no impact on the analysis of the determinants of engagement which we carry out in the next section.

Add Figures 11 and 12 about here

-----

# 4.2 Factors influencing user activity and user engagement

In this section, we focus on the determinants of engagement and analyse them separately. More specifically, we consider the overall population of posts generated by Italian regional DMOs in 2013 and analyse how they relate to the following variables: a) type of post; b) length of the post; c) posting time of day; d) posting day of the week; e) posting month of the year; f) frequency of posting.

*Type of post.* As noted previously, most of the Italian regional DMOs tend to post mainly photos. Our analysis shows that photos and videos are also the type of content that elicits most comments (Figure 13).

Add Figure 13 about here

*Length of post.* The analysis on the overall population of posts shows that the majority of comments are between 100 to 400 characters long, with posts of around 200 characters being the most frequent (they appear in 30% of cases):

Add Figure 14 about here

\_\_\_\_\_

It also emerges that the length of the post can have an impact on engagement, with posts between 100 and 200 characters in length (which are incidentally the most frequent) generating the peaks of engagement (Figure 15).

# Add Figure 15 about here

-----

*Posting time of day.* The analysis on the overall population of posts also shows that the majority of DMO posts are made in the morning or afternoon (ordinary working hours), although several also post overnight.

-----

Add Figure 16 about here

------

Content posted in the evening seems to have a positive effect on generic engagement, probably because such posts are fewer in number (and therefore more visible), and also because users can more easily access Facebook in the evening (see Figure 17).

-----

Add Figure 17 about here

*Posting day of the week.* The analysis shows that the majority of DMO posts are made during the working week (Monday to Friday), with just 17% of DMO content being posted during the weekend. It is difficult to identify a clear-cut relationship between the day of the week when content is posted and the resultant engagement. That said, it does seem that content posted over the weekend elicits

higher engagement (Figure 18).

Add Figure 18 about here

0

It appears that DMO posting activities and the responses to those posts by users are asymmetric: DMOs post mainly during the working week whereas users interact more during the weekend.

*Frequency of posting.* In 28% of cases, DMOs generate less than one post per day, whereas in 10% of the cases DMOs post more than 6 posts a day (Figure 19).

Add Figure 19 about here

\_\_\_\_\_

The effect of posting frequency on engagement is illustrated in Figure 20, which shows that maximum engagement occurs when the daily frequency is low. In other words, the lower the frequency of posting, the higher the engagement. This can be interpreted to mean that overwhelming Facebook users with too many posts is counterproductive, and does not command higher engagement.

Add Figure 20 about here

\_\_\_\_\_

-----

*Posting month of the year.* The empirical analysis shows that, on average, DMOs post the most during the months of May, July and September, whereas activity is lowest in April and August.

Engagement, on the other hand, is highest in January, after which it drops significantly in February, and then continues declining smoothly until May. Thereafter it begins to rise again, reaching a relative peak in July, before starting to fall off again during the autumn. The January peak might be explained by the visits of tourists interested in winter holidays (or busy with Christmas-holiday post-trip assessments), while the July peak may be ascribed to people preparing for their summer trips (August, when most Italian businesses close, is the month of highest domestic tourism flows). However there seems not to be any clear-cut relationship between the month of posting and the engagement metrics.

The variables presented in this section are summarised in Table 8, which also shows the correlation between these variables and engagement.

Add Table 8 about here

\_\_\_\_\_

Based on the observations made thus far, concerning the relationships between individual variables and engagement, we formulate the following research hypotheses:

H1: The type of posts made by a regional DMO has an impact on engagement; more specifically, visual types of posts (such as photos) have a positive impact on engagement
H2: High frequency of posting by a regional DMO has a negative impact on engagement
H3: The length of the posts made by a regional DMO has a positive impact on engagement
H4: Evening posts by a regional DMO have a positive impact on engagement

These hypotheses were validated using multivariate ordinary least squares (OLS) regressions. OLS regression is a generalised linear modelling technique that may be used to model a single response variable on a number of different independent variables. OLS regression proves to be particularly useful when variables display multivariate normality, as in the case of our study (highest skewness .44 for weekend posts, higher kurtosis .60 for morning posts). The following variables were included in the model: *Engagement* was used as the dependent variable, regressed against *Type\_ photo, Type\_status, Post frequency, Average length of post, Evening posts, Morning posts, Week-end posts* used as independent variables. Additionally, two control variables were included, namely *Posting month* (each month was first included as an explanatory variable, and then a dummy was created including just those months identified as having the relative peaks of engagement: i.e. January, July, and November) and *Geography* (four dummy variables coded as South; Centre; North; Islands). Table 9 shows the results of the regression analysis.

Add Table 9 about here

The proposed model was found to have a good fit, explaining 31% of the variance. Considering the proposed hypotheses: H1 cannot be rejected (with 10% statistical significance); H2 and H3 likewise cannot be rejected (with 1% statistical significance); whereas H4 and H5 generate results that are not statistically significant. Thus, the regression results suggest that Italian regional DMOs increase their engagement if they post less frequently, and if they select mainly visual content (such as photos), possibly accompanied by moderately long text. Posting during the weekend positively affects engagement, but not to a statistically significant extent. By contrast, posting in the morning has a significant negative impact on engagement-possibly simply because DMOs are most active during the morning, while their Facebook fans are much less active at that time.

The significant link between visual content and engagement confirms that tourists (or potential travellers) who use online social networks are attracted by images of the sights and destinations that they have visited or plan to visit. This is consistent with the strong relationship between tourism and image-making media related to sightseeing traditionally featured in tourism studies (e.g. Beeton, 2004), and the fact that DMOs recognise tourists' preference for visual content sharing (Munar and Jacobsen, 2014). As far as post length is concerned, the positive link between moderately long posts and engagement could be explained by the fact that when the DMOs task is to disseminate

information they produce narrative content rich in details: this behaviour seems symmetric with the behaviour that tourists adopt when producing narrative content (Stoeckl, Rohrmeier, & Hess, 2007).

The negative impact of posting frequency on engagement is likely ascribable to user's time constraints. Frequent posting implies sharing larger amount of content during a given day/week. However, Internet users will not have enough time during a given day/week to engage with all the content posted within that time frame. Moreover, since users tend to look only at the latest content available on a Facebook page, any older posts they have not yet seen will anyhow be displaced by the newer ones, and so receive fewer comments, likes or shares, thereby resulting in lower engagement.

Morning posts likely have a negative impact on engagement because Facebook users seem to spend more time on DMO Facebook pages during the evening and at night. Since users react mainly to the more recent posts, this will favour content that has been posted in the evening. These results are in line with statistics about the usage of the Facebook platform, which show that most users access Facebook during the early or late evening (Vitrue, 2010).

The control variable "geography" has a positive impact on engagement, but with a higher magnitude for the Centre and the Islands, where several of the most active DMOs (i.e., Tuscany, Emilia Romagna, Marche, and Sicily) are located. Finally, the month of posting seems to have no influence on engagement.

# 4.3 Social media strategies employed by Italian regional DMOs: interaction with consumers and users

Interviews conducted with relevant key informants at 17 of the 20 Italian regional DMOs revealed that, to some extent, Italian regional DMOs use social media simply to advertise and promote in a conventional way. However several DMOs (such as Trentino and Emilia Romagna) believe that social platforms should be used as a form of social narration, rather than as just another channel for advertising. In the words of Giovanna Arata, community manager for the Emilia-Romagna Facebook page:

"Social media in general, and Facebook in particular, must be a platform for a collective expression and narration, not just another medium through which to serve advertising"

This reveals what appears to be a divide. On one side, a significant number of regions use social media (Facebook, Twitter, etc.) simply as just another advertising and marketing channel, changing little about the content of the message (in other words, the ads and messages displayed on the official website are essentially replicated on the Facebook pages and Twitter accounts). On the other side, a small number of regions are instead using social media in a more experimental way, to communicate differently and

with differentiated content. Among these, we can identify Trentino, Tuscany, Sicily, Marche, Liguria, and Emilia-Romagna.

As emerges clearly from the extensive analysis of posts/content conducted on the overall population of Italian regional DMOs' Facebook pages, the regions with the highest proportion of posts by users (Trentino, Tuscany and Sicily, see Table 7) are also among those that manage to leverage visual content in a more creative and experimental way.

Generally speaking the content analysis shows that, in 2013, the posts which generated more than 100 comments were 142 in total, distributed as follows: 86 (60.6%) for Trentino, 26 (18.3%) for Tuscany, 19 (13.4%) for Sicily, 6 (4.2%) for Liguria, 3 (2.1%) for Abruzzo and 1 (0.7%) for both Puglia and Emilia-Romagna. Of these, the type of post which generated the highest engagement in terms of comments was photos: 138 (97.2%), followed by status changes 4 (2.8%). Most of the pictures were associated with opinion polls, quizzes, or games. Table 10 summarises the features of the top ten posts, in terms of number of comments, in 2013:

Add Table 10 about here

As Table 10 clearly shows, the managers of the Tuscany Facebook page often organise opinion polls asking users which of the presented photos/images they would like as a cover (see Picture 1). This proves to be an excellent strategy for involving and engaging users, as borne out by the number of Likes (1,263) related to the picture.

Add Picture 1 about here

\_\_\_\_\_

Another technique frequently used by several Italian DMOs is quizzes. In Picture 2, the marketers of Visit Trentino ask for the meaning of a term in Trentino dialect (i.e., *prosac*, a kind of white wine known as *prosecco*), inviting users to submit their answers through comments. Users who know the answer will be stimulated to respond, thereby raising the visibility of the Facebook page and prompting other users to be involved in the quiz.

\_\_\_\_\_

21

# Add Picture 2 about here

\_\_\_\_\_

As nearly all the interviewees asserted (and the analysis of the posted content confirms) images have an unparalleled power for marketing a destination through Facebook, and when this power is leveraged through opinion polls, quizzes, and games, the capability to attract Likes, Comments, and Shares, and thereby boost user engagement, is significantly increased. Trentino, Tuscany, and Sicily are certainly destinations that have understood this point very well.

# 4.4 Facebook and topics

The analysis shows that most of the Italian regional DMOs use social platforms (in this case Facebook) to convey a wide array of information to their users. The information relayed on social platforms covers a variety of topics that have a bearing on tourism. For example, the Tuscany DMO focuses on three main thematic areas: arts, food and wine, and 'around Tuscany' (which collects travel and general information). The Emilia Romagna DMO includes content related to the Adriatic coast (especially during the summer), art and culture, nature, well-being and sports, motor valley, and wine and food. The Marche DMO organises its information into clusters centred on different specific types of tourism (rather than tourism in general) and tourism experiences (rather than tourism destinations).

The strategic objectives of regional DMOs for social-media development vary widely. For example, the Emilia-Romagna DMO is investing in further internationalisation (at present the content is available in Italian, English, and German, and is being developed in Portuguese and Spanish as well). It is also planning to differentiate its accounts based on topic areas (i.e., wine and food, motor valley, arts and culture), and trying to increasingly integrate its online and offline activities, for example by organising visits to museums, excursions, bike tours, family events, or photographic walks, often linked to online contests.

The Tuscany regional DMO, as underlined by Costanza Giovannini, social media manager for Fondazione Sistema Toscana, is developing a strategy based on the main brand "Tuscany", which is present on all the digital platforms and is also used on other websites such as "Voglio vivere cosi – Turismo.inToscana.it" on Facebook, and "Diari Toscani" on Twitter.

The Marche regional DMO, the only one that has a presence on all 10 most important social networks for tourism (Facebook, Twitter, Flickr, Pinterest, Instagram, Forsquare, Panoramio, Google+, YouTube and Issuu) is working to develop the region's brand reputation through the involvement of 'influencers', whose online activity can become 'viral'. Moreover, Mr. Giorgetti, social media strategist of the Marche DMO underlines that:

"The style used should be friendly, politically correct, colloquial and sympathetic, but simultaneously also kind of authoritative because it comes from a public organisation".

# 4.5 DMOs' organisational structure and social media

The roles and job titles of the professionals who manage DMOs' social media efforts vary significantly across organisations. For example, the Emilia-Romagna DMO manages its web marketing strategy through a community manager, Giovanni Arata, and 5 editors. Of these, a total of 3 people work specifically on social media (2 full-time and 1 part-time).

In the Tuscany DMO, 18 people work in the web marketing department. The original social media team consisted 12 people and was outsourced until mid-2012 to another company, called Fondazione Sistema Toscana. Since then, the team has been internalised and now counts 10 members (9 full-time, and 1 part-time). Of these, 1 is the head of web marketing and communication, 1 is a social media analyst, 2 are social media strategists, 5 are content managers, and 1 is a technology specialist.

The Veneto DMO has a total of 5 people dealing with its web presence, of which 2 work on social media: one full-time, and the second part-time.

The Social Media Team of the Marche DMO consists of 1 coordinator/team leader (the social media strategist) and 4 more members. All of them also work on social media.

In the Valle D'Aosta DMO, the web marketing team consists of 4 people (1 web marketing strategist, 2 community managers and 1 editor<sup>7</sup>) who are permanently employed in the web marketing department: of these, 2.5 people work on social media (meaning that two people are totally dedicated to social media while a third works part-time on social media).

For the Friuli-Venezia-Giulia DMO, 4 permanent employees are in charge of web marketing, but only 2 of them work part-time directly on social media. The web marketing team consists of 1 web marketing coordinator and strategist, while the other 3 professionals are editors (with a varying degree of specialisation).

Within the Abruzzo DMO, 7 permanent employees are in charge of web marketing: all of them work part-time on social media. The Abruzzo region has recently hired 2 more people as content managers. Overall there is not a clear cut specialisation in terms of roles and job titles: everyone deals with a number of tasks.

<sup>&</sup>lt;sup>7</sup> The community managers and the editor work on content so they could be defined as content managers as well.

For the Basilicata DMO, web marketing and social media are dealt with by an external company consisting of 1 person working full-time as web coordinator, and another working part-time (as content manager). Just one of these two people works part-time on social media

Within the Lombardy DMO, 8 permanent employees work on web marketing; of these, 3 work on social media (2 full-time and 1 part-time). For the Piedmont DMO, 5 permanent employees deal with web marketing: of these, 1 person works full-time on social media. In the Sicilian DMO, 2 permanent employees work on web marketing and 1 works full-time on social media. Surprisingly, the representative of the Campania DMO did not respond to our questions.

Taken together, the above empirical evidence reveals some interesting features and trends: a) web and social media marketing activities are very rarely outsourced; b) in general, only employees who work full-time on web marketing are permanent employees, and they typically deal with strategic and coordination tasks related to web marketing; c) for more specific tasks (related for example to content generation) young professionals are typically hired with temporary contracts on specific projects<sup>8</sup>, which in some cases are later transformed into permanent contracts. The data collected are summarised in Table 11:

-----

Add Table 11 about here

The above results have to be interpreted with caution for a number of reasons. First, most of the staff working on web marketing do not necessarily dedicate 100% of their time to marketing tasks, but typically deal with administrative duties as well. This is especially true for permanent employees of the DMO. Additionally, it is often not easy to distinguish how much a specific staff member works on social media, as they might handle social media along with other web marketing activities.

#### 4.6 Funding

Information about financial resource allocation to regional DMOs' web marketing was extremely difficult to collect. Unfortunately, many DMOs replied that either the information was too confidential to be shared, or that they simply did not have a specific break-down that would allow them to identify exactly what share of their promotional budget is spent on web marketing. Table 12 summarises the data collected through the interviews.

<sup>&</sup>lt;sup>8</sup> The most frequently used form of contract is the CO.CO.PRO. (Contratto di Collaborazione a Progetto), which basically implies that the employee is hired for a limited number of months to work on a specific project.

#### Add Table 12 about here

-----

Since most of the money that regional DMOs spend on promotional activities is public money levied at the regional level, it is rather surprising that this information is in many cases not disclosed. Also, even those figures that have been disclosed must be interpreted with caution. The direct costs related to web marketing are relatively easy to identify when the human resources devoted to web marketing are shortterm employees hired specifically for that task. On the other hand, such costs are extremely difficult to determine when social media and web marketing are managed by permanent employees (because permanent staff typically also perform also other tasks). Furthermore, some DMOs include expense items for personnel wages within their web marketing and social media marketing budgets, while other do not. Finally, even for staff directly employed in a DMO's web marketing department, it is in many cases difficult to quantify how much time they devote exclusively to social media.

Despite the aforementioned methodological issues, the collected data seem to indicate that the regions that allocate more money to web marketing and social media are also those which appear to be more effective and experimental in their approaches to the web (see the case of Tuscany).

## 4.7 Audience development

The empirical evidence of this study shows that the Italian regional DMOs which most successfully increased their Facebook fan base in 2013 were Umbria and Calabria—although both started from a very thin fan base (around 100 fans for each). However, among the DMOs with the largest fan bases, Tuscany and Sicily recorded significant increases (up 50% during the year, thanks also to the massive investment in social media, in the case of Tuscany, as is apparent from the section on funding).

There is significant variance across organisations in terms of strategies for audience development. As yet, no DMO uses its social platforms for commercial purposes. For example, Turismo Emilia Romagna (TER) is investing substantially in social media because it believes this can enable it to reach a wider audience, particularly segments that cannot be reached through more traditional media. The community manager of TER, Giovanni Arata says that:

"Social media are not yet being used for commercial purposes because users do not perceive social media as an area in which to carry out transactions" This philosophy is shared by Costanza Giovannini, social media manager of the Tuscany DMO, who is however planning to create an ad-hoc page for commercial purposes. As far as audience development is concerned, Mrs. Giovannini maintains that:

"We do not deploy any specific strategy to increase the number of fans on Facebook (or followers on Twitter), but we try to focus our efforts on call to action on Twitter. We believe that, to capture the attention of Internet users, it is crucial to use the right tone of voice and language, with an emphasis on emotional aspects rather than commercial ones."

In line with this approach, the Tuscany DMO is starting up a new project aimed at telling the story of Tuscany through people, rather than through places. In April 2013 this DMO also started a project called Play Your Tuscany, consisting of a blog tour for bloggers, photographers, and video professionals, who are invited to participate in themed itineraries and recount their travel experiences through photos, stories, and videos on their blogs.

The Marche DMO has never engaged in advertising activities, and its number of fans is a result of a social strategy that was implemented mainly through the involvement of influencers--as is clear from the words of its social media strategist. There is no direct use of social media for commercial purposes.

# 4.8 Success in social media

How do we measure the success of the social media strategies crafted and deployed by Italian regional DMOs? The preceding analysis shows that engagement outcomes can vary significantly depending on how the engagement metrics are constructed. If we deploy metrics normalised by the number of fans on a monthly basis (see Table 3) and on a rolling basis, the best performers are: Calabria, Umbria, and Veneto (Figure 8) for the monthly basis; and Umbria, Calabria, and Abruzzo for the rolling basis. However, if we instead use metrics not normalised by the number of fans, the picture changes and Trentino, Tuscany, and Marche are the best performers (see Figure 10). This distinction is a significant contribution of this paper, which helps to highlight how engagement metrics can potentially be misleading, when it is unclear how those metrics are constructed.

Although the literature on social media ROI is rapidly developing (see Fisher, 2009; Frick, 2010), the results of the quantitative content analysis, combined with the qualitative data collected through interviews with DMOs, suggest that there is no general agreement on a single dominant metric for gauging success in social media strategies.

More specifically, the interviews highlighted that there is no specific set of metrics deployed consistently across organisations to assess the success of social media. Indeed, many interviewees said that their regional DMO is not particularly interested simply in the number of Likes or followers garnered on social media. On the other hand, during 2013 many DMOs adopted Facebook Ads to help promote their pages. As pointed out by Mr. Arata from the Emilia Romagna DMO, the Return On Investment (ROI) of social media strategies for the Emilia Romagna regional DMO is measured through a combination of different tools: Blogmeter and Radian 6 (paid service) and Google Analytics, Facebook Insights, Pinreach (free service).

The Tuscany DMO, in evaluating the success of its social media campaigns, also takes into account the level of engagement. According to Mrs. Giovannini, its social media manager, it is more important to notice how people "react" to a Facebook post in terms of sentiment, rather than merely deploying "cold numbers" to assess the effectiveness of social media strategies.

In addition, none of the interviewed social media managers are measuring commercial outcomes: that is, to what extent their strategies on Facebook and other social media are increasing the number of tourism packages and services purchased at the destination. They say that this is far beyond their capabilities, since the majority of DMO sites are set up mainly for promotion, and do not directly sell tourism products and services.

In any event, many DMO managers do not seem particularly concerned with the monetary ROI of their Facebook page, because they consider public money spent on e-promotion to be anyhow well spent (this looks like an a priori assumption for many of them). Nevertheless, greater transparency is needed concerning the way promotional budgets are allocated by individual DMOs, since it is Italian taxpayers' money that Italian regional DMOs are spending for their promotional activities.

# 5. Conclusions and implications

The results of this analysis clearly show that each Italian regional DMO has its own distinctive manner of using and interpreting social media strategies on Facebook. At first sight, one might (inaccurately) conclude there is a 'social media divide' between the north and south of Italy, since the top 5 DMOs in terms of number of Facebook posts (Trentino, Emilia Romagna, Marche, Tuscany, and Liguria) are all located in northern and central Italy. However, looking more carefully at the engagement data and the managerial practices in place, a more fine-grained distinction emerges, between regional DMOs that use/interpret social media in an active and experimental fashion vs. those that use/interpret social media in a more passive way. The former group includes Trentino, Tuscany, Sicily (which is nowhere near northern or central Italy), Emilia Romagna, and Marche—all of which are striving to develop effective social media strategies aimed at enhancing engagement and interaction with Internet users. These same regions also tend to be present on a number of additional social media platforms, such as YouTube, Pinterest, Flickr, and Foursquare.

On the passive side are several DMOs (typically those of smaller regions such as Molise) that use social media merely to share information and news that has already bee published on the DMO's official website or YouTube channel, without involving Internet users. Moreover, minor regions such as Molise are present on Facebook with just an ordinary profile (rather than a fan page), which entails a number of limitations. For example, the number of friends is limited to 5,000, and to be able to follow the profile users must send a 'friend request' to the account (in the case of Molise, the "Molisani nel mondo" account). Several DMOs have given their Facebook page the same name as the region (see the Liguria DMO), which helps make the Facebook page highly recognisable (with a potentially positive impact on the number of fans).

Our findings indicate that visual content (namely photos) and moderately long posts have a statisticallysignificant positive impact on DMOs' Facebook engagement metrics, whereas higher post frequency and early daily timing (in the morning) of posts have a negative impact on engagement. The strong link between visual content and engagement may be ascribed to the well-established relationship between tourism and image-making media (e.g. Beeton, 2004), and the prevalence of images in DMOs posts also shows that DMOs recognise tourists' preference for visual content sharing (Munar and Jacobsen, 2014). The fact that moderately long posts boost engagement may be ascribed to the fact that, when the DMOs disseminate information they produce narrative content rich in details: this behaviour seems symmetric with the behaviour that tourist adopt when producing narrative content (Stoeckl, Rohrmeier, & Hess, 2007).

High posting frequency has a negative impact on engagement mainly because, owing to time constraints, there are limits to the amount of content that Internet users can absorb or engage with on any given day or week. Also, since users tend to consider only the newest content, older posts which they have missed will not receive an equal number of comments, likes or shares actions.

Posting in the morning has a negative impact on engagement because—as statistics on the usage of the Facebook platform show (Vitrue, 2010)--Facebook users tend to view these pages in the evening and at night, and so they will react more to the newest posts (those which were posted in the evening).

By triangulating the statistical analysis and interviews conducted in this study, we can arrive at some managerial implications for Italian regional DMOs. First, it seems that regions such as Trentino, Tuscany, and Sicily have developed some managerial best practices in the social media arena that could be usefully leveraged by other, less adept regions, which currently merely replicate on social media the content posted on their official websites. Such strategies include creating opinion polls, games, quizzes,

and contests involving visual content (e.g., to select the cover picture of the Facebook page). Our results clearly indicate that it would benefit other DMOs—also in other geographical contexts, and at any level (country, region, local)--to strategically deploy visual content to elicit their users' engagement. Given tourists' known preference for visual content sharing (Munar and Jacobsen, 2014), this approach is likely to boost social activity on their pages. Additionally, embedding such visual content in opinion polls or quizzes seems to produce an even stronger 'call to action' for users, that enhances social activity even more. Yet, as this analysis shows, only a few social media managers are currently aware that the hedonic value arising from UGC might be used to influence users' travel-planning decisions (Ayeh, Norman & Law, 2013) or to strengthen the reputation of their destination by creating positive electronic word-of-mouth (Dijkmans et al., 2015; Luo and Zhong, 2015).

Second, it seems that the apparent digital divide between the north and south of Italy--which anyhow exists only for total posts--is decreasing over time as many regions in the south (and also the islands of Sicily and Sardinia) recognise social media as an effective marketing tool. Indeed, a major culture shift is in progress, facilitated by national initiatives such as Buy Tourism Online (BTO), the conference and trade show on web marketing for tourism held annually in Florence (now preparing its 8th edition), which is attended by a growing number of Italian destination marketers. The general improvement in digital culture is a goal that regional DMO managers must pursue within their organisations, to help them face up to ever-increasing global competition among national and domestic destinations. In fact, the larger DMOs (such as Tuscany, Trentino, and Emilia-Romagna) do this already, and are trying to keep up the good work by increasing their social media budgets. However the smaller DMOs have only recently realised the importance of social media strategies, and started allocating a greater share of their promotional budgets and staff to social media in general and to Facebook in particular, and investing more in training their employees on digital marketing. This finding, concerning the importance of building a strong digital culture, might be transferable to other DMOs around the globe. Specifically, what can make a difference in increasing the social engagement of a DMO Facebook page is the ability of DMO employees to keep up to date with the latest developments in social media marketing. This in turn is tightly related to the DMO managers' attitude to digital culture, and their awareness of how social media can contribute to build a positive destination image (Frías et al., 2012; Llodra-Riera et al., 2015). The managerial best practices observed in this work can be regarded as simply the natural outcome of a new strategic management philosophy (which is not necessarily widespread among the analysed DMOs) that prioritises investments in social media and in the professional development of the marketing staff.

Third, Italian regions generally still deploy social media in a traditional top-down manner instead of exploring ways to strengthen a bottom-up UGC-driven approach and better interact and engage with

users/consumers. This might be due to the fact that UGC is simply less interesting (or less professionally produced) than content generated by DMOs themselves. Only a small number of Italian DMOs (such as Trentino and, to a certain extent, Tuscany and Emilia-Romagna) are trying to incorporate social media into an overarching flexible strategy, that allows for experimentation and for enhancing engagement and informal conversation. These results seem to corroborate the findings of a pioneering study on the use of social media by national DMOs (Hays et al. 2013) and suggest that that the approach of Italian DMOs to Web 2.0 fails to follow Kaplan and Haenlein's (2009) list of five key actions for managing a social media presence: be active, be interesting, be humble, be informal, and be honest. Overall, juxtaposing social media with other marketing tools might pave the way for Italian regional DMOs have much room for improvement and should benchmark themselves against other comparable DMOs in other parts of the world.

Fourth, it is still unclear how DMOs' social media managers tackle the issue of measuring the success and effectiveness (and efficiency) of their social media and Facebook activity. While it is clear that the largest regional DMOs have made important investments in improving their companies' digital culture, they still lack some understanding of the instruments that could be used to assess their efforts in web marketing in general, and in social media in particular. Administrators of individual Facebook accounts use different metrics (such as Facebook Insights and Google analytics) to assess their social media performance, but there is huge variance across social media managers in the way these metrics are adopted and made sense of.

Fifth, while a number of metrics and Key Performance Indicators (KPIs) are referred to in day-to-day operational activities (Socialbakers, 2015) and are used in practice, they very rarely inform strategic marketing management decisions. The reasons for this are twofold. On one hand, it is extremely difficult to assess the exact investment (in terms of financial and, especially, human resources) allocated solely to social media (separating social media marketing from other digital marketing activities). For instance, even if a metric for Social Media ROI could be theoretically agreed upon by managers and tourism management researchers, calculating it in practice would require having comparable data for the financial and human resources deployed by the social media teams of all the DMOs. On the other hand, any benchmarking exercise against competing destinations and DMOs is not feasible, because individual Facebook administrators do not share their information with competitors. To our knowledge, this is the case for a number of DMOs worldwide.

The software tool developed for this study was presented to most of the interviewed Italian DMO social media managers (covering 85% of the population of regional DMO social media managers nationwide). According to those managers, this software represents a first, crucial step toward a solid benchmarking exercise based on public data, complemented with a reliable and robust algorithm for calculating social engagement. They also read with interest a draft version of this paper, and noted that it provides comparative evidence and a knowledge base useful for learning more about their social media presence and benchmarking themselves with their competitors.

This study has a number of limitations. First, the analysis was carried out in a specific geographical context (Italy), where tourism promotion is handled mainly by a specific level of government (i.e. the regions) rather than by the central government. Consequently, this study can only be generalised to countries where regional levels of government, or individual municipalities, play a predominant role in destination marketing (this is the case of confederate states in federal states, see for example the Länder in Germany). These findings could be further generalised by enlarging the sample to include regional and municipal DMOs in other countries. Furthermore, it might be interesting to benchmark our findings against other European destinations in Germany, the Scandinavian countries, Spain, and the UK, and by also taking National Tourism Organisations into account. A second, limit is that this analysis of Facebook content and DMO strategies and tactics was carried out over one recent year (2013). In future work the time frame of the analysis could be expanded to include several more years in the past, or updated with more recent data. Another way of extending this analysis would be to include additional specifications for the regression model, by adding such explanatory variables as the number of employees working on social media or the ratio of the social media budget to the overall web marketing budget. In addition, we are currently working on integrating a sentiment analysis module into our software tool, since the currently available tools suffer from several methodological weaknesses and generally work mostly for the English language. This development might add further nuances to our analysis. Finally, it would be interesting to explore whether--for destinations with comparable degrees of natural and cultural attraction--those DMOs that invest more and adopt managerial best practices for social media are effectively able to derive from this a sustained competitive advantage for their destinations.

#### REFERENCES

Alfaro, I., Nardon, M., Pianesi, F., Stock, O., & Zancanaro, M. (2005). "Using cinematic techniques on mobile devices for cultural tourism", *Information technology & Tourism*, 7(2), 61–71.

Ayeh, J.K., Au, N., Law, R, (2013). "Predicting the intention to use consumer-generated media for travel planning", *Tourism Management* 35 (2013) 132-143.

Baggio, R., (2014). "Technological Innovation in e-Tourism: the role of interoperability and standards". In M.M Mariani, R. Baggio, D. Buhalis and C. Longhi (Eds.), *Tourism Management, Marketing and Development: The Importance of Network and ICTs* (pp. 41–55). New York: Palgrave-Macmillan.

Baggio R., Czakon, W., & Mariani, M.M., (eds.), (2013). Managing Tourism in a Changing World: Issues and Cases, London: Routledge..

Baloglu, S., & Pekcan, Y. A. (2006) "The website design and Internet site marketing practices of upscale and luxury hotels in Turkey", *Tourism Management*, 27(1), 171–176.

Beeton, S. (2004). "Rural tourism in Australia: has the gaze altered? Tracking rural images through film and tourism promotion", *Progress in Tourism and Hospitality Research*, 6(3), 125-135.

Berger, S., Lehmann, H., & Lehner, F., (2003). "Location-based services in the tourist industry", Information Technology & Tourism, 5(4), 243-256.

Bilgihan, A. Barreda, a., Okumus, F. Nusair, K. (2016). "Consumer perception of knowledge-sharing in travelrelated Online Social Networks", *Tourism Management* 52, 287-296.

Boyd, D., & Ellison, N., (2008). "Social network sites: Definition, history, and scholarship". *Journal of Computer-Mediated Communication*, 13, 210–230.

Borges, B. (2009). Marketing 2.0: Bridging the gap between seller and buyer through social media marketing. Tucson, AZ: Wheatmark.

Buhalis, D. & Foerste, M. (2015). "SoCoMo Marketing for Travel and Tourism: empowering co-creation of value", Journal of Destination Marketing & Management, forthcoming

Buhalis, D., & Law, R., (2008). "Progress in tourism management: Twenty years on and 10 years after the internet: The state of eTourism research", *Tourism Management*, 29(4), 609-623.

Buhalis, D., & Licata, M. C., (2002). "The future eTourism intermediaries", Tourism Management, 23(3), 207-220.

Buhalis, D., & O'Connor, P., (2005). "Information communication technology-Revolutionising tourism", *Tourism Recreation Research*, 30(3), 7–16.

Brás, J.M.; Costa, C., & Buhalis, D., (2010). "Networks analysis and wine routes: the case of the Bairrada wine route, Portugal", *The Services Industries Journal*, 30(10), 1–21.

Chung, J.Y., & Buhalis, D., (2008). "Information needs in online social networks", Information Technology and Tourism, 10(4), 267-282.

Clemons, E. K., Hann, I.-H., & Hitt, L. M., (2002). "Price dispersion and differentiation in online travel: An empirical investigation", *Management Science*, 48(4), 534–549.

D'Aveni, R., (1994). Hypercompetition: managing the dynamics of strategic manoeuvring. New York: The Free Press.

Dijkmans, C., Kerkhof, P., Beukeboom, C.J. (2015). "A stage to engage: Social media use and corporate reputation", *Tourism Management* (47): 58-67.

Dominici, L., (2013). "Viaggi. Boom dei social network", Il Sole 24 Ore, 1 Marzo 2013.

Feng, R., Morrison, A.M., & Ismail, J.A., (2003). "East versus West: A comparison of online destination marketing in China and the USA", *Journal of Vacation Marketing*; 10(1). 43-56.

Fiore, A. M., Kim, J., & Lee, H., (2005). "Effect of image interactivity technology on consumer responses toward the online retailer". *Journal of Interactive Marketing*, 19(3), 38–53.

Fisher, T. (2009). "ROI in social media: A look at the arguments". Database Marketing & Customer Strategy Management, 16(3), 189-195.

Flouri, E., & Buhalis, D. (2004). "Wireless technologies for tourism destinations". In A. J. Frew (Ed.), *Information and communication technologies in tourism* 2004 (pp. 27–38). New York: Springer Wien.

Fotis, J., Buhalis, D., & Rossides, N., (2011). "Social media impact on holiday travel: The case of the Russian and the FSU markets", *International Journal of Online Marketing*, 1(4), 1-19.

Frías, D.M., Rodríguez, M.A., Alberto Castañeda, J., Sabiote, C.M., & Buhalis, D, (2012). "The Formation of a Tourist Destination's Image via Information Sources: The Moderating Effect of Culture", *International Journal of Tourism Research*, 14(5), 437-450.

Frick, T. (2010). Return on Engagement: Content, Strategy, and Design Techniques for Digital Marketing. Burlington, MA: Focal Press.

Gelb, B. D., & Sundaram, S., (2002). "Adapting to 'word of mouse", Business Horizons, 45(4), 15-20.

Gioia, D. A., Corley, K. G., & Hamilton, A. G. (2013). "Seeking qualitative rigor in inductive research: notes on the Gioia methodology", *Organization Research Methods*, 16(1), 15-31.

Go, F. M., & Williams, A. P., (1993). "Competing and co-operating in the changing tourism channel system", *Journal of Travel and Tourism Marketing*, 2(2/3), 229–248.

Gretzel, U., (2006). "Consumer generated content – trends and implications for branding". *E-review of Tourism* Research, 4(3), 9–11.

Gretzel, U., Yuan, Y., & Fesenmaier, D., (2000). "Preparing for the new economadvertising strategies and change in destination marketing organizations", *Journal of Travel Research*, 39, 149–156.

Gursoy, D., & McCleary, K., (2004). "An integrative model of tourists' information search behaviour", Annals of Tourism Research, 31(2), 353-373.

Hashim, N. H., Murphy, J., & Law, R., (2007). "A review of hospitality website design frameworks". In M. Sigala, L. Mich, and J. Murphy (Eds.), *Information and communication technologies in tourism* 2007 (pp. 219–230). New York: Springer Wien.

Hays, S., Page, S., & Buhalis, D., (2013). "Social media as a destination marketing tool: its use by national tourism organisations", *Current Issues in Tourism*, 16 (3), 211-239.

ISTAT (2015). Annuario Statistico Italiano 2014. Roma: ISTAT

Jakkilinki, R., Georgievski, M., & Sharda, N., (2007). Connecting destinations with an ontology-based e-tourism planner. In M. Sigala, L. Mich, & J. Murphy (Eds.), Information and communication technologies in tourism 2007 (pp. 21–32). Springer Wien.

Jick, T. D. (1979). "Mixing qualitative and quantitative methods: Triangulation in action", Administrative Science Quarterly, 24, 602-611.

Kaplan, A.M., & Haenlein, M., (2009). "Users of the world, unite! The challenges and opportunities of social media", *Business Horizons*, 53, 59-68.

Leung, D., Law, R., van Hoof, H. & Buhalis, D., (2013). "Social media in tourism and hospitality: a literature review", *Journal of Travel & Tourism Marketing*, 30, 3–22.

Llodra-Riera, Isabel, Martínez-Ruiz, M.P., Jimenez-Zarco, A.J., Izquierdo-Yusta, A. (2015). "A multidimensional analysis of the information sources construct and its relevance for destination image formation", *Tourism Management* 48, 319-328.

Luo, M., Feng, R., & Cai, L. A., (2004). "Information search behavior and tourist characteristics: The Internet vis-a`-vis other information sources", *Journal of Travel & Tourism Marketing*, 17(2/3), 15–25.

Luo, Q. & Zhong, D. (2015). "Using social network analysis to explain communication characteristics of travelrelated electronic word-of-mouth on social networking sites", *Tourism Management* 46 274-282.

Matloka, J., & Buhalis, D., (2010). "Destination Marketing through User Personalised Content (UPC)", in Gretzel, U., Law, R., Fuchs, M., (eds), ENTER 2010 Proceedings, Lugano, Springer-Verlag, Wien, ISBN: 9783211994061, pp.519-530.

Michopoulou, E., Buhalis, D., Michailidis, S., & Ambrose, I., (2007). "Destination management systems: Technical challenges in developing an eTourism platform for accessible tourism in Europe". In M. Sigala, L. Mich, & J. Murphy (Eds.), *Information and communication technologies in tourism* 2007 (pp. 301–310). Springer Wien.

Miles, M. B. & Huberman, A. M. (1984). Qualitative Data Analysis: A Sourcebook of New. Methods. California; SAGE publications

Moore, K. and McElroy, J.C. (2012), "The influence of personality on Facebook usage, wall postings, and regret", *Computers in Human Behavior*, 28(1), 267-274.

Munar, A.M., & Steen Jacobsen, J.K. (2014). "Motivations for sharing tourism experiences through social media", *Tourism Management*, 43, 46-54.

Niininen, O., Buhalis, C., & March, R., (2007). "Customer empowerment in tourism through Consumer Centric Marketing (CCM)", *Qualitative Market Research*, 10(3), 265-282.

Yovcheva, Z., Buhalis, D., & Gatzidis, C., (2012). "Smartphone Augmented Reality Applications for Tourism", e-Review of Tourism Research (eRTR), 10(2), 2012

Odinma, A. C., Oborkhale, L. I., & Kah, M. M. O., (2007). "The trends in broadband wireless network technologies", *The Pacific Journal of Science and Technology*, 8(1), 118–125.

O'Connor, P., & Frew, A., (2002). "The future of hotel electronic distribution: Expert and industry perspectives", *Cornell Hotel and Restaurant Administration Quarterly*, 43(3), 33–45.

Oviedo-García, A., Muñoz-Expósito, M., Castellanos-Verdugo, M. & Sancho-Mejías, M. (2014). "Metric proposal for customer engagement in Facebook", *Journal of Research in Interactive Marketing*, 8(4), 327-344

Parra-Lopez, E., Bulchand-Gidumal, J., Gutierrez-Tano, D. & Diaz-Armas, R. (2011). "Intentions to use social media in organizing and taking vacation trips", *Computers in Human Behavior*, 27(2), 640-654.

Patton, M.Q. (1990). Qualitative Evaluation and Research Methods, 2nd ed., Sage, Newbury Park, CA.

Pike, S, and Page, S. (2014). Destination Marketing Organizations and destination marketing: A narrative analysis of the literature, *Tourism Management*, 41, 202-227.

Qualman, E. (2009). Socialnomics: How social media transforms the way we live and do business. Hoboken, NJ: John Wiley and Sons.

Raggam, K., & Almer, A., (2005). "Acceptance of geo-multimedia applications in Austrian tourism organizations". In M. Sigala, L. Mich, and J. Murphy (Eds.), Information and communication technologies in tourism 2007 (pp. 163–174). New York: Springer Wien.

Rheingold, H., (1993): The virtual community: Homesteading on the electronic frontier. Reading, MA: Addison-Wesley.

Ricci, F., & Werthner, H., (2006). "Recommender systems", International Journal of Electronic Commerce, 11(2), 5-9.

Ruiz-Mafe, C., Martí-Parreño, J. and Sanz-Blas, S. (2014), "Key drivers of consumer loyalty to Facebook fan pages", Online Information Review, 38(3), pp. 362-380

Sabate, F., Berbegal-Mirabent, J., Cañabate, A. and Lebherz, P.R. (2014), "Factors influencing popularity of branded content in Facebook fan pages", *European Management Journal*, 32(6): 1001-1011.

Shea, L., Enghagen, L., & Khullar, A., (2004). "Internet diffusion of an e-complaint: A content analysis of unsolicited responses", Journal of Travel & Tourism Marketing, 17(2/3), 105–116

Schegg, R., Liebrich, A., Scaglione, M., & Ahmad, S.F.S., (2008). "An exploratory field study of web 2.0 in tourism". *Information and Communication Technologies in Tourism*, 5, 152–163.

Schetzina, C., (2010): Introduction to social media analytics. New York, NY: PhoCusWright.

Sigala, M., Christou, E., & Gretzel, U. (2012). Social Media in Travel, Tourism and Hospitality: Theory, Practice and Cases. Ashgate Publishing, Ltd.

Social Bakers (2015). Social Bakers Analytics. User Guide. Available online at: http://analytics.socialbakers.com/template/classic/media/SocialbakersAnalyticsGuide.pdf

Stabb, S., & Werthner, H., (2002). "Intelligent systems for tourism", IEEE Intelligent Systems, November December, 53-55.

Stankov, U., Lazic, L., & Dragicevic, V., (2010). "The extent of use of basic Facebook user-generated content by the national tourism organizations in Europe", *European Journal of Tourism Research*, 3(2), 105–113.

Stepchenkova, S., Mills, J. E., & Jiang, H., (2007). "Virtual travel communities: Self-reported experiences and satisfaction". In M. Sigala, L. Mich, & J. Murphy (Eds.), *Information and communication technologies in tourism* 2007 (pp. 163–174). New York: Springer Wien.

Stoeckl, R., Rohrmeier, P., & Hess, T. (2007). Motivations to produce user generated content: differences between webbloggers and videobloggers. In 20th Bled eConference eMergence: Merging and emerging technologies, processes, and institutions (pp. 398e413).

Tufte, B., & Rasmussen, J., (2003). "Children on the net-State of the art and future perspectives regarding Danish children's use of the Internet", *European Advances in Consumer Research*, 6, 142–146.

UNWTO (2015). Tourism Highlights, 2015 Edition. Madrid: UNWTO.

Vogt, C. A., & Fesenmaier, D. R., (1998). "Expanding the functional information search model", Annals of Tourism Research, 25(3), 551-578.

Wang, K.-C., Hsieh, A.-T., Yeh, Y.-C., & Tsai, C-W., (2004). "Who is the decision-maker: The parents or the child in group package tours?", *Tourism Management*, 25(2), 183–194.

Wang, Y., Yu, Q., & Fesenmaier, R. D., (2002). "Defining the virtual tourist community: Implications for tourism marketing", *Tourism Management*, 23(4), 407–417.

Wang, Y., & Fesenmaier, D. R., (2004). "Towards understanding members' general participation in and active contribution to an online travel community", *Tourism Management*, 25(6), 709–722.

Weber, R.P., (1990). Basic content analysis (2nd ed.). Newbury Park, CA: Sage Publications.

Wenger, A., (2008). "Analysis of travel bloggers' characteristics and their communication about Austria as a tourism destination", *Journal of Vacation Marketing*, 14, 2: 169-176.

Werthner, H., & Ricci, F., (2004). "E-Commerce and tourism", Communications of the ACM, 47(12), 101-105.

White, N., & White, P., (2007). Home and away: Tourists in a connected world. *Annals of Tourism Research*, 34(1), 88–104.

Xiang, Z., & Gretzel, U., (2010). "Role of social media in online travel information search". *Tourism Management*, 31, 179–188.

# Websites

Turismo Emilia Romagna: "Logiche di promozione 2.0 dei luoghi", www.slideshare.net/turismoemiliaromagna/

Facebook Graph API 2.0 (2014): https://developers.facebook.com/blog/post/2014/04/30/the-new-facebook-login/

Vitrue (2010): When are Facebook users most active? http://mashable.com/2010/10/28/facebook-activity-study/#FIatHlejkkqN